Author's response to reviews

Title: Return-to-work of sick-listed workers without an employment contract - what works?

Authors:

Sylvia J. Vermeulen (s.vermeulen@vumc.nl)
Sietske J. Tamminga (s.j.tamminga@amc.uva.nl)
Antonius J.M. Schellart (ton.schellart@vumc.nl)
Jan Fekke Ybema (jan_fekke.ybema@tno.nl)
Johannes R. Anema (h.anema@vumc.nl)

Version: 7 Date: 9 July 2009

Author's response to reviews:

Dear editor,

After reading the submitted final manuscript one more time, one of the co-authors (he had been on vacation earlier) noticed correctly that we refer in our manuscript to loglinear regression (this was added after a comment of one of the reviewers). However, this is incorrect and has to be logistic regression. Also, I found two more typographical errors after reading the manuscript once more. I have made corrections regarding the logistic regression analysis and the typographical errors.

Kind regards,

Sylvia Vermeulen
Insurance physician/researcher
VU medical Center, EMGO institute
Amsterdam, the Netherlands